Author's response to reviews

Title: The Validation of a Needed Measure of Multiethnic Older Women's Social Quality of Life: Two Cross-Sectional Studies

Authors:

   Luciana Lagana` (LLASPPHI@aol.com)
   Maria L. Bratly (mariabrately@ymail.com)
   Ioakim Boutakidis (iboutakidis@exchange.fullerton.edu)

Version: 5 Date: 27 August 2011

Author's response to reviews:

August 26, 2011

Dear Dr. Neilan,

I am detailing below the three changes that I have made to the manuscript, with the help of my co-authors. I hope that you will find them satisfactorily.

Structure: We have modified the manuscript’s structure as required; consequently, we have made several edits throughout the manuscript.

Tables: We have repositioned some of the values within the tables.

Tables: Table 7 is now one table (in which the results of our two studies are reported side-by-side).

Thank you very much for your support.

Best wishes,

Luciana Lagana`